For reprint orders, please contact: reprints@futuremedicine.com

Isatuximab for relapsed/refractory multiple myeloma: review of key subgroup analyses from the Phase III ICARIA-MM study

Paul G Richardson^{*,1}, Simon J Harrison², Sara Bringhen³, Fredrik Schjesvold^{4,5}, Kwee Yong⁶, Frank Campana^{‡,7,8}, Solenn Le-Guennec, Sandrine Macé & Meletios A Dimopoulos ¹Medical Oncology, Dana-Farber Cancer Institute, Boston, MA 02215, USA

²Peter MacCallum Cancer Centre & Royal Melbourne Hospital Melbourne, Sir Peter MacCallum Department of Oncology,

University of Melbourne Parkville, Victoria, 3010, Australia

³Myeloma Unit, Division of Hematology, University of Torino, Azienda-Ospedaliero Universitaria Città della Salute e della Scienza di

Torino, Torino, 10126, Italy

⁴Oslo Myeloma Center, Oslo University Hospital, Oslo, 0188, Norway

⁵KG Jebsen Center for B-Cell Malignancies, University of Oslo, Oslo, 0315, Norway

⁶Department of Haematology, University College London Hospitals, London, WC1N 3BG, UK

⁷Sanofi R&D, Cambridge, MA 02142, USA

⁸Currently, Takeda Pharmaceuticals, Cambridge, MA 02142, USA

9Sanofi R&D, Vitry-sur-Seine 94400, France

¹⁰Department of Clinical Therapeutics, National & Kapodistrian University of Athens, Athens 157 72, Greece

*Author for correspondence: Tel.: +1 617 632 2127; Paul_Richardson@dfci.harvard.edu

[‡]Affiliation at the time of study

In the Phase III ICARIA-MM study (Trial Registration number: NCT02990338 [ClinicalTrials.gov]), the addition of the anti-CD38 monoclonal antibody isatuximab to pomalidomide and dexamethasone led to increased progression-free survival and improved response rates in patients with relapsed/refractory multiple myeloma. There is an unmet treatment need, particularly among patients with poor prognoses, including those with high-risk cytogenetics, those who have renal impairment, those who are elderly and those who are refractory to prior lines of treatment. In this review, the subgroup analyses from the ICARIA-MM study, representing subpopulations with poor prognostic factors, are discussed. Overall, the addition of isatuximab to pomalidomide and dexamethasone improved progression-free survival and disease response rates across different subgroups, regardless of prognostic factor.

Lay abstract: Currently, the majority of patients with multiple myeloma are not cured, and current treatments may not be helpful for patients with poor prognoses, including those with high-risk chromosomal changes, those who have impaired kidney function, those who are elderly and those who are refractory to prior treatments. In this review, we will discuss the benefits of the combination of isatuximab plus pomalidomide and dexamethasone in these difficult-to-treat patients.

Tweetable abstract: Isatuximab combination improves progression-free survival and disease response rates in difficult-to-treat patients with multiple myeloma.

First draft submitted: 6 May 2021; Accepted for publication: 16 August 2021; Published online: 15 September 2021

Keywords: anti-CD38 monoclonal antibody • elderly patients • high-risk cytogenetics • ICARIA-MM • isatuximab • multiple myeloma • prior lines • refractory • relapsed • renal impairment

Multiple myeloma (MM) is a neoplasm of B-cell lineage characterized by the proliferation of malignant plasma cells in the bone marrow, with consequent anemia and increased bone resorption by osteoclasts, often resulting in bone fractures or hypercalcemia [1]. The malignant plasma cells produce and secrete immunoglobulin (Ig), usually of the IgG class, which is seen as a monoclonal component (M-protein) on serum electrophoresis. M-protein and



Future

JCOLOG

free light chains, part of the Ig, accumulate in the bloodstream and urine of patients with MM, and commonly induce renal failure [2].

MM is the second most frequent hematologic malignancy after lymphoma, totaling 1% of all cancers and approximately 10% of all hematologic malignancies [3,4]. In 2020, the global age-adjusted annual incidence of MM was 1.8/100,000, with a mortality rate of 1.1/100,000 [5]. Furthermore, there were approximately 176,404 new cases of MM across the world and more than 117,077 people died from the disease in 2020. MM typically affects elderly patients, with a median age at diagnosis of 70 years; only 37% of patients diagnosed with MM are younger than 65 years of age [6].

Currently, the majority of patients with MM are not cured; however, the use of autologous stem cell transplantation as intensification therapy and the use of consolidation therapy in patients with newly diagnosed MM has prolonged median progression-free survival (PFS) [7,8]. The availability of proteasome inhibitors (PIs; bortezomib, carfilzomib and ixazomib) and immunomodulatory agents (thalidomide, lenalidomide [Len] and pomalidomide) has significantly extended the survival of patients with MM [9–11]. However, patients with MM continually relapse over time, highlighting an urgent need for new therapeutic approaches [12]. Monoclonal antibodies such as daratumumab (anti-CD38) and elotuzumab (antisignaling lymphocytic activation molecule family member 7 [SLAMF7]) were approved for the treatment of relapsed/refractory MM (RRMM) in 2015 [13], and isatuximab (anti-CD38) was approved in 2020 for patients with two prior therapies, including Len and a PI [14].

Historically, in the USA and Europe, the most common first- and second-line therapies for MM include bortezomib and Len-based regimens, respectively [15–17]. These regimens, with or without the glucocorticoid dexamethasone, represented the most common approaches in the earlier 2010s, with expansion into triplet therapy with newer agents (e.g., carfilzomib, ixazomib, daratumumab and elotuzumab) in 2015 [15,16]. Despite the introduction of targeted therapies and combination regimens, patients with MM continue to experience multiple relapses and/or become refractory to treatment [18]. Outcomes, including duration of response and overall survival (OS), worsen with successive lines of therapy [19].

There are several identified factors for MM that can portend prognosis and may be important in terms of determining treatment decisions. Such prognostic factors are categorized by disease burden, host factors, tumor biology and depth of response to therapy [20]. Worse outcomes have been observed in patients with high LDH, increased plasma cell proliferative activity, and in those who have been classified as International Staging System stage III. Additional subgroups of patients with MM and a poor prognosis include those with high-risk cytogenetics, those who have renal impairment, those who are elderly and those who are refractory to prior lines of treatment. This review summarizes data from the Phase III ICARIA-MM trial (isatuximab in combination with pomalidomide and dexamethasone in RRMM) analyzed by each of these four subgroups that negatively impact prognosis.

Isatuximab, a monoclonal antibody targeting CD38

CD38 is a transmembrane glycoprotein involved in the regulation of migration, signal transduction and receptormediated adhesion [21–23]. Furthermore, CD38 serves as an ectoenzyme with cyclic ADP-ribose hydrolase activity, catalyzing the metabolism of calcium messengers [23]. CD38 is highly expressed in MM cells and at relatively low levels in other tissues, making this a desirable target for therapeutic antibodies to treat MM [21].

Isatuximab is a monoclonal antibody that binds to a specific epitope of CD38 [24,25]. Isatuximab has anti-MM activity via multiple biological mechanisms (Figure 1). Some mechanisms are mediated by the Fc portion of the antibody, which binds to the Fc γ receptors expressed on effector natural killer cells and macrophages to trigger antibody-dependent cell-mediated cytotoxicity and antibody-dependent cellular phagocytosis, or that allows the fixation of the human complement to trigger complement-dependent cytotoxicity [25,26]. Unlike other anti-CD38 antibodies, isatuximab is able to induce direct apoptosis in the absence of cross-linking agents and to inhibit CD38 ectoenzyme activity, which may be specific to the CD38 epitope targeted by isatuximab [24,27]. Increasing CD8⁺ T-cell responses against NY-ESO-1 were observed in two patients who exhibited an autologous IgG and IgM antibody response against this antigen; these patients also exhibited a newly developed CD8⁺ T-cell response against the antigen; these patients [28]. Therefore, tumor-specific immune fitness may be associated with clinical responses, and the development of adaptive antitumor immunity in response to isatuximab treatment is clinically relevant.

This review will summarize the data found in the pivotal Phase III ICARIA-MM trial, focusing on the results for specific high-risk patient subgroups, including high-risk cytogenetics, renal impairment, elderly patients and number of prior lines and refractory status.



Figure 1. Mechanisms of action of isatuximab. The CD38 antibody isatuximab was shown to kill multiple myeloma cells by ADCP, ADCC, CDC via the MAC, and direct apoptosis. Isatuximab also inhibits CD38 ectoenzyme activity and the production of immune-suppressive adenosine, which likely alleviate immunosuppressive microenvironment in bone marrow niche of multiple myeloma patients.

ADCC: Antibody-dependent cellular cytotoxicity; ADCP: Antibody-dependent cellular phagocytosis; CDC: Complement-mediated cytotoxicity; MAC: Membrane attack complex. Copyright 2018 by Sanofi [29].

Phase III trial of isatuximab in combination with pomalidomide in RRMM patients (ICARIA-MM)

ICARIA-MM (NCT02990338) was a prospective, randomized, open-label, active-controlled, multicenter Phase III study that compared treatment of isatuximab in combination with pomalidomide and dexamethasone ([Pd], Isa-Pd) with Pd [30,31].

In ICARIA-MM, 307 patients who had received ≥ 2 prior lines of therapy including Len and a PI were enrolled. Patients in the Isa-Pd arm received 10 mg/kg of isatuximab intravenously weekly for the first 4 weeks, then every other week thereafter. Both arms received approved doses of Pd every 28 days. Treatment continued until disease progression or unacceptable toxicity. The primary end point was PFS. Secondary end points included overall response rate (ORR) and OS.

Randomized patients (154 Isa-Pd and 153 Pd) had a median age of 67 (range: 36–86) years and 19.5% had high-risk cytogenetics. Patients received a median of 3 (range: 2–11) prior lines of therapy, with 92.5% being refractory to Len and 75.9% being refractory to PIs.

At a median follow-up of 11.6 months, PFS was 11.5 months (95% CI: 8.9-13.9) in the Isa-Pd arm versus 6.5 months (95% CI: 4.5-8.3) in the Pd arm, with a hazard ratio (HR) of 0.60 (95% CI: 0.44-0.81); p = 0.001.

The ORR was 60.4% in Isa-Pd versus 35.3% in Pd (p < 0.0001). The proportion of patients who achieved at least a very good partial response (VGPR) rate or better was 31.8% in the Isa-Pd versus 8.5% in the Pd arm. The median time to progression was longer in the Isa-Pd group (12.7 months; 95% CI: 11.2–15.2) than in the control group (7.8 months; 95% CI: 5.0–9.8).

Grade ≥ 3 adverse events (AEs) were observed in 86.8% of patients in the Isa-Pd arm compared with 70.5% in the Pd arm. However, only 7.2% versus 12.8% of patients discontinued treatment due to AEs in the Isa-Pd and Pd arms, respectively. The most frequent treatment-emergent AEs (TEAEs; any grade) were infusion reactions ([IRs] 38.2% Isa-Pd vs 0% Pd), and respiratory infections (28.3% Isa-Pd vs 17.4% Pd). The most frequent laboratory

abnormality was neutropenia, being reported in 24.3% (grade 3) and 60.5% (grade 4) in the Isa-Pd arm versus 38.8% (grade 3) and 31.3% (grade 4) in the Pd arm.

Overall, the addition of isatuximab to pomalidomide and dexamethasone resulted in a significant and clinically meaningful benefit in PFS and response rate in patients with RRMM, with a manageable safety profie [30,31]. Importantly, the results of ICARIA-MM formed the basis for the approval of isatuximab. Isatuximab (Sarclisa[®]) is approved in a number of countries in combination with pomalidomide and dexamethasone for the treatment of adult patients with RRMM who have received at least two prior therapies, including Len and a PI. Based on the Phase III IKEMA study, isatuximab in combination with carfilzomib and dexamethasone is approved in the USA for the treatment of adult patients with RRMM who have received one to three prior lines of therapy, and in the European Union for the treatment of adult patients with MM who have received at least one prior therapy [14,32–36].

ICARIA-MM subgroup information

Overall, the ICARIA-MM study population was representative of the global population with RRMM, including important subgroups of patients with characteristics associated with poor prognosis: high-risk cytogenetics defined by the presence of del(17p), translocation t(4;14) and/or translocation t(14;16) (19.5%), renal impairment defined as creatinine clearance <60 ml/min/1.73 m² as determined using the Modification of Diet in Renal Disease (MDRD) equation (36.2%), 75 years of age or older (19.9%), International Staging System stage III (25.1%), Eastern Cooperative Oncology Group performance score of 2 (10.4%) and high levels of LDH (32.2%).

Importantly, all patient subgroups benefited from the addition of isatuximab [31], as shown in Table 1 [31,37-40] and further discussed in the next sections of this review.

ICARIA-MM subgroup: high-risk cytogenetics

A prespecified subgroup analysis of the ICARIA-MM study assessed whether the PFS benefit with isatuximab in addition to pomalidomide and dexamethasone is maintained in patients with high-risk cytogenetics versus standard-risk patients [37,41].

Risk classification based on cytogenetic profiling and careful analysis of risk subgroups is becoming increasingly important in the evaluation of novel MM therapies [42]. Some of the genetic abnormalities that have been associated with MM include the following: translocations of chromosome 14 (t[4;14], t[11;14], t[14;16] and t[14;20]); amplification of chromosomal region 1q21.3; deletion of chromosomal region 17p; and mutations in *KRAS*, *NRAS*, *FGFR3* and *TP53* [12,43]. The International Myeloma Working Group classifies the presence of t(4;14), t(14;16), t(14;20), del(17p) and/or gain(1q21) as high-risk cytogenetic markers associated with reduced survival of patients with MM [18,44.45]. Patients who have these abnormalities are considered high-risk, whereas standard-risk patients are characterized by the absence of such abnormalities [44–46]. Importantly, high-risk cytogenetic patients have a median OS of only 3 versus 10 years in standard-risk cytogenetics patients.

In ICARIA-MM, the cytogenetic risk was characterized by central laboratory fluorescence *in situ* hybridization testing of purified CD138⁺ plasma cells from baseline bone marrow aspirate, which was obtained and interpretable in 241 (78.5%) patients [31]. Missing results in 66 (21.5%) patients were either because samples were not provided for the test or due to poor sample quality.

High-risk cytogenetic status was defined by the presence of at least one of the following chromosomal abnormalities: del(17p), translocation t(4;14) or translocation t(14;16). An abnormality was considered positive if it was present in at least 30% of analyzed plasma cells, except for del(17p) where the threshold was at least 50%. Sensitivity analyses were conducted to assess the impact of the cytogenetic cutoff definition used.

High-risk chromosomal abnormalities were present in fewer patients in the Isa-Pd treatment arm (15.6%) than in the Pd control arm (23.5%), with del(17p) and t(4;14) being the most frequent abnormalities in both arms. A total of 12.1% (9.1% Isa-Pd, 15% Pd) of the patients had del(17p), and 8.5% (7.8% Isa-Pd, 9.2% Pd) had t(4;14). Overall, eight (2.6%) patients (1.9% Isa-Pd, 3.3% Pd) had two high-risk chromosomal abnormalities that included del(17p) and either t(4;14) or t(14;16); this is a patient population with a very poor prognosis.

Among high-risk patients, PFS events were reported in 14/24 (58.3%) in the Isa-Pd group compared with 22/36 (61.1%) in the Pd group. Compared with a PFS of 3.7 months with Pd alone, the PFS with the addition of isatuximab was 7.5 months (HR: 0.66; 95% CI: 0.33–1.28). Although median PFS was lower in high-risk patients (7.5 months) compared with standard-risk patients (11.6 months) receiving Isa-Pd, the benefit of isatuximab was similar to that observed in standard-risk patients (HR: 0.62; 95% CI: 0.42–0.93). High-risk patients with t(4;14) receiving Isa-Pd exhibited prolonged median PFS compared with those receiving Pd (7.5 vs 2.8 months; HR:

Table 1. Summar	y of safet	y and effica	cy subgro	up analyses	from ICARIA	N-MM.						
	PFS	(months)		ORR (%)	Any TE	AE, n (%)†	Grade ≥3	TEAE, n (%) [†]	Serious 1	rEAE, n (%)†	TEAE leadi treatment (ng to definitive liscontinuation, (%) [†]
	Isa-Pd	Pd	Isa-Pd	Pd	lsa-Pd	Pd	Isa-Pd	Pd	Isa-Pd	Pd	Isa-Pd	Pd
Overall ICARIA-MM population	11.5	6.5	60.4	35.3	151 (99.3)	146 (98.0)	132 (86.8)	105 (70.5)	94 (61.8)	80 (53.7)	11 (7.2)	19 (12.8)
High-risk cytogenetics												
– High risk	7.5	3.7	50.0	16.7	23 (100)	32 (94.1)	22 (95.7)	23 (67.6)	17 (73.9)	17 (50.0)	2 (8.7)	8 (23.5)
– Standard risk	11.6	7.4	65.0	42.3	102 (99.0)	75 (98.7)	88 (85.4)	58 (76.3)	60 (58.3)	47 (61.8)	7 (6.8)	6 (7.9)
Renal impairment												
- <45 ml/min/1.73 m ²	7.5	2.8	35.0	23.5	I	I	I	I	I	I	I	I
- <60 ml/min/1.73 m ²	9.5	3.7	56.4	24.5	54 (100)	47 (100)	49 (90.7)	37 (78.7)	42 (77.8)	28 (59.6)	6 (11.1)	7 (14.9)
$- \ge 60 \text{ ml/min/1.73 m}^2$	12.7	7.9	67.8	42.7	85 (98.8)	91 (96.8)	74 (86.0)	63 (67.0)	44 (51.2)	48 (51.1)	5 (5.8)	11 (11.7)
Elderly												
– ≥75 years	11.4	4.5	53.1	31.0	32 (100)	28 (100)	30 (93.8)	21 (75.0)	22 (68.8)	16 (57.1)	5 (15.6)	4 (14.3)
– 65–74 years	11.6	8.6	64.7	38.9	66 (100)	52 (98.1)	56 (84.8)	40 (75.5)	41 (62.1)	32 (60.4)	2 (3.0)	8 (15.1)
- <65 years	11.5	5.0	59.3	34.3	53 (98.1)	66 (97.1)	46 (85.2)	44 (64.7)	31 (57.4)	32 (47.1)	4 (7.4)	7 (10.3)
Prior lines and refractory	y status											
- 2 prior lines	12.3	7.8	57.8	35.6	I	I	I	I	I	I	I	I
- 2-3 prior lines	12.3	7.8	56.9	38.6	45 (100)	43 (97.7)	39 (86.7)	28 (63.6)	27 (60.0)	21 (47.7)	1 (2.2)	3 (6.8)
$- \ge 3$ prior lines	11.4	5.9	61.5	35.2	106 (99.1)	103 (98.1)	93 (86.9)	77 (73.3)	75 (62.6)	59 (56.2)	10 (9.3)	16 (15.2)
- >3 prior lines	9.4	4.3	67.3	28.8	I	I	I	I	I	I	I	I
- 4 prior lines	8.5	3.3	56.3	28.6	I	I	I	I	I	I	I	I
– Len refractory	11.4	5.6	59.0	31.4	I	I	I	I	I	I	I	I
- Pl refractory	11.4	5.6	60.2	32.2	I	I	I	I	I	I	I	I
– Len + Pl refractory	11.2	4.8	58.6	29.9	I	I	I	I	I	I	I	I
 Refractory to Len at last line 	11.6	5.7	55.9	29.5	I	I	I	I	I	I	I	I
[†] Refractory status is a refle Isa-Pd arm	ection of treatn	nent resistance of	the myeloma c	lone and should no	ot influence system	iic safety. TEAE in	cidence should b	e interpreted in li	ght of the differe	ence in treatment	duration due to	longer PFS in the
d: Dexamethasone; Isa: Isa	tuximab; Len: .	Lenalidomide; ORF	R: Overall resp.	onse rate; P: Pomali	domide; PFS: Prog	Iression-free survi	val; PI: Proteason	ne inhibitor; TEAE	: Treatment-eme	rrgent adverse ev	ent.	

0.49; 95% CI: 0.19–1.31). The isatuximab treatment effect on PFS was less pronounced in high-risk patients with del(17p) (9.1 vs 7.4 months; HR: 0.76; 95% CI: 0.30–1.92). There was no significant interaction at the 10% level between cytogenetic risk status and treatment groups. There was an observed PFS benefit in both high- and standard-risk patients treated with Isa-Pd, which was less pronounced in patients with del(17p).

Compared with Pd alone, Isa-Pd treatment led to improved ORR from 16.7 to 50.0% in patients with high-risk cytogenetics. A VGPR or better rate was observed in 29.2% of patients in the Isa-Pd cohort versus 2.8% in the Pd cohort, and the treatment benefit was maintained regardless of the high-risk cytogenetic cutoff definition used.

The median time-to-progression benefit observed with isatuximab in the overall ICARIA-MM population was maintained in patients with high-risk cytogenetics. The HRs for del(17p) were 0.62 (95% CI: 0.25–1.53) at 5% cutoff and 0.68 (95% CI: 0.19–2.48) at 60% cutoff. For t(4;14), HRs were 0.34 (95% CI: 0.11–1.04) at 3% cutoff and 0.14 (95% CI: 0.03–0.67) at 60% cutoff. These results indicate that the time-to-progression benefit with isatuximab was maintained in patients with high-risk cytogenetic abnormalities, and the benefit was independent of the cutoff definition used.

Although cross-trial comparisons must be approached with caution due to differences in patient populations, subgroup analyses of other studies investigating anti-CD38 monoclonal antibody combinations in patients with high-risk cytogenetics have demonstrated PFS benefit and increased depth of response. In the Phase III POLLUX study, patients receiving daratumumab plus lenalidomide and dexamethasone exhibited prolonged PFS compared with lenalidomide and dexamethasone alone in both standard- (HR: 0.43; 95% CI: 0.32-0.57) and high-risk (HR: 0.34; 95% CI: 0.16-0.72) patients [47]. Results from a subgroup analysis of the Phase III CASTOR study demonstrated prolonged PFS in patients with standard (HR: 0.26; 95% CI: 0.19-0.37) and high (HR: 0.41; 95% CI: 0.21-0.83) cytogenetic risk treated with daratumumab, bortezomib and dexamethasone compared with bortezomib and dexamethasone alone [48]. In POLLUX and CASTOR, the threshold for positive findings was determined locally and varied by site for each chromosomal abnormality, whereas there were consistent thresholds used by central laboratory analysis in ICARIA-MM, which may contribute to more reliable results. Additionally, POLLUX did not include patients who were refractory to or intolerant of lenalidomide, whereas ICARIA-MM included these patients. Results of an exploratory analysis from the EQUULEUS study investigating the combination of daratumumab, pomalidomide and dexamethasone demonstrated a median PFS of 10.3 months (95% CI: 4.6-NE) for patients with standard cytogenetic risk and 3.9 months (95% CI: 2.3-NE) in those with high cytogenetic risk [49]. This study did not have a control group, so it is difficult to determine the impact of daratumumab on the PFS of patients with standard versus high cytogenetic risk; however, patients with high cytogenetic risk had lower PFS. In the ICARIA-MM study, the benefit of isatuximab in high-risk patients was similar to that observed in standard-risk patients.

Patients with gain(1q21)

A retrospective analysis of plasma cells revealed that 128 of 307 (41.7%) patients had gain(1q21) abnormality, with 26.1% having three copies of gain(1q21) and 15.6% having \geq 4 copies. Overall, 85 of 307 (27.7%) patients had isolated gain(1q21), which was defined as those patients who had no other high-risk chromosomal abnormalities, including del(17p), t(4;14) or t(14;16), and more patients treated with isatuximab had isolated gain(1q21) (36.4 vs 19.0%). Compared with Pd alone, longer median PFS was observed in patients treated with Isa-Pd who had isolated gain(1q21) (11.2 vs 4.6 months; HR: 0.50; 95% CI: 0.28–0.88) and in patients without gain(1q21) and standard risk (15.2 vs 9.8 months; HR: 0.64; 95% CI: 0.29–1.38). Patients treated with Isa-Pd also exhibited a higher ORR compared with Pd-treated patients, regardless of gain(1q21) status. Compared with patients who received Pd, a higher proportion of patients who received Isa-Pd achieved VGPR or better, including those with isolated gain(1q21) (25.0 vs 3.4%) and those without gain(1q21) with standard risk (34.5 vs 11.4%).

Results from a subgroup analysis of the Phase III ELOQUENT 2 study demonstrated that patients with gain(1q21) treated with the combination of elotuzumab, lenalidomide and dexamethasone exhibited a reduced risk for disease progression or death [50]; however, the combination of elotuzumab, pomalidomide and dexamethasone did not contribute to a PFS benefit in this patient subgroup in the ELOQUENT-3 study [51]. A meta-analysis of daratumumab studies demonstrated a survival benefit with the addition of daratumumab in standard- and high-risk patients; however, the impact of gain(1q21) was not discussed [52]. A recent study investigating the prognostic impact of gain(1q21) in patients who received daratumumab demonstrated shorter PFS and OS among those with gain(1q21) (PFS: 0.5 years [95% CI: 0.3–1.4]; OS: 0.9 years [95% CI: 0.7–2.3]) than patients without gain(1q21) (PFS: 2.1 years [95% CI: 1.9–not reached]; OS: not reached [95% CI: 1.9 years–not reached]). To date, these

findings do not demonstrate a treatment benefit of daratumumab among patients with gain(1q21), whereas the ICARIA-MM study demonstrates improved PFS and ORR with isatuximab treatment in this difficult-to-treat patient subgroup [53].

Safety

Isatuximab had a manageable safety profile in both high-risk and standard-risk patients. More grade \geq 3 AEs occurred in patients who received Isa-Pd compared with Pd who had high-risk cytogenetics (95.7 vs 67.6%) or standard-risk cytogenetics (85.4 vs 76.3%). The incidence of serious adverse events (SAEs) was also higher following isatuximab treatment compared with Pd alone in the high-risk subgroup (73.9 vs 50.0%), but not in the standard-risk group (58.3 vs 61.8%). The addition of isatuximab to Pd did not increase treatment discontinuations in the high-risk (8.7% Isa-Pd vs 23.5% Pd) or standard-risk subgroups (6.8% Isa-Pd vs 7.9% Pd). Furthermore, treatment-related mortality did not increase in either subgroup following isatuximab treatment compared with the control group.

The most frequent TEAEs in the high-risk Isa-Pd group versus the control group were febrile neutropenia (13 vs 0%) and pneumonia (21.7 vs 17.6%). IRs of grade \geq 3 occurred only in the Isa-Pd group, at a frequency of 8.7% in the high-risk population and 1% in the standard-risk population. Among high-risk patients, G-CSF was used in 16 (28.1%) and 21 (36.8%) patients treated with Isa-Pd or Pd, respectively. Among standard-risk patients, G-CSF was used in 70 (39.1%) and 35 (19.6%) patients treated with Isa-Pd or Pd, respectively. Overall, these data demonstrate that the addition of isatuximab to pomalidomide and dexamethasone provides a benefit over pomalidomide and dexamethasone therapy alone in the difficult-to-treat subgroup of patients with RRMM who have high-risk cytogenetics, consistent with the results from the overall ICARIA-MM study population.

ICARIA-MM subgroup: renal impairment

Renal impairment affects up to 50% of patients with MM, mainly due to the accumulation and precipitation of Ig free light chains in the distal tubules, resulting in tubule obstruction and cast neuropathy [54]. Moreover, renal impairment is an independent predictor of adverse survival outcomes for MM patients, and is associated with a higher rate of AEs, early mortality and reduced OS [55]. The median survival of patients with renal impairment is approximately half that of patients with no renal impairment [56]. Entry criteria in most clinical trials result in the exclusion of many patients with renal dysfunction; therefore, enrolled candidates may not be representative of the typical patient with MM [50,57–59]. Notably, in the ICARIA-MM, patients were included if they had baseline estimated glomerular filtration rate (eGFR) \geq 30 ml/min/1.73 m², determined using the MDRD equation [31].

An ICARIA-MM subgroup analysis was conducted to examine the efficacy of Isa-Pd in patients with moderate renal impairment (eGFR \geq 30 to <60 ml/min/1.73 m²) [38]. In ICARIA-MM, patients were required to have an eGFR \geq 30 ml/min/1.73 m² [31]. An eGFR was calculated using the MDRD equation at screening; within 24 h prior to study treatment administration on days 1, 8, 15 and 22 of cycle 1; within 24 h prior to study treatment administration on day 1 of every subsequent cycle; at the end of treatment visits; and as clinically indicated. eGFR results were classified as renal impairment (<60 ml/min/1.73 m²) or no renal impairment (\geq 60 ml/min/1.73 m²); data were also examined among patients with eGFR <45 ml/min/1.73 m². Complete renal response was defined as an improvement in eGFR less than 50 ml/min/1.73 m² at baseline to \geq 60 ml/min/1.73 m² in at least one post-baseline assessment, in accordance with the International Myeloma Working Group recommendations [60]. Responses were considered durable (sustained) when lasting \geq 60 days. A minor renal response was defined as an improvement in eGFR from less than 15 ml/min/1.73 m² at baseline to \geq 15 to less than 30 ml/min/1.73 m², or from \geq 15 to less than 30 ml/min/1.73 m² at baseline to \geq 30 ml/min/1.73 min² in at least one assessment during treatment.

In the Isa-Pd arm, the number of patients with renal impairment at baseline was 39% with eGFR less than 60 ml/min/1.73 m² and 38% with eGFR \geq 30 to 60 ml/min/1.73 m². In the Pd arm, 34% of patients had eGFR less than 60 ml/min/1.73 m² and 33% had eGFR \geq 30 to 60 ml/min/1.73 m². No patients were receiving renal replacement therapy or hemodialysis.

The median PFS for patients with renal impairment was 9.5 months in patients receiving Isa-Pd (n = 55) and 3.7 months in those receiving Pd (n = 49; HR: 0.50; 95% CI: 0.30–0.85). Among patients with an eGFR less than 45 ml/min/1.73 m², the median PFS was 7.5 with Isa-Pd (n = 20) and 2.8 months with Pd (n = 17; HR: 0.50; 95% CI: 0.22–1.13). Among patients without renal impairment, median PFS was 12.7 months in the Isa-Pd arm (n = 87) and 7.9 months in the Pd arm (n = 96; HR: 0.58; 95% CI: 0.38–0.88).

At the time of the primary analysis, median OS for patients with renal impairment was not reached in patients receiving Isa-Pd versus 11.6 months in those receiving Pd (HR: 0.53; 95% CI: 0.30–0.96). Among patients with an eGFR less than 45 ml/min/1.73 m², median OS was 10.7 months for the Isa-Pd arm versus 6.6 months for the Pd arm (HR: 0.62; 95% CI: 0.26–1.45). Among patients without renal impairment, median OS was not reached with Isa-Pd or Pd (HR: 0.62; 95% CI: 0.33–1.19).

A higher ORR was achieved with Isa-Pd versus Pd treatment, regardless of renal impairment status. The ORR was 56.4% (Isa-Pd) and 24.5% (Pd) for patients with renal impairment (odds ratio [OR]: 3.98; 95% CI: 1.60–10.17). Among those with renal impairment, a VGPR or better was achieved by more patients treated with Isa-Pd (32.7%) than Pd (4.1%). A total of eight patients receiving Isa-Pd achieved MRD negativity (sensitivity level: 10^{-5}); of these, three had an eGFR less than 60 ml/min/1.73 m². No patients receiving Pd achieved MRD negativity. Among patients with eGFR less than 45 ml/min/1.73 m², the ORR was 35.0 and 23.5% in the Isa-Pd and Pd arms, respectively (OR 1.75; 95% CI: 0.34–10.11). Among patients without renal impairment, the ORR was 67.8% in the Isa-Pd arm and 42.7% in the Pd arm (OR: 2.83; 95% CI: 1.48–5.42). Among patients with mild renal impairment (eGFR \geq 45 to <60 ml/min/1.73 m²), the ORR was 68.6% in those treated with Isa-Pd (n = 35), which was similar to the ORR in patients without renal impairment. For patients with mild renal impairment in the Pd arm (n = 32), the ORR was 25.0%.

Complete renal responses were observed in 71.9% (23/32) of patients receiving Isa-Pd and 38.1% (8/21) of those receiving Pd (OR: 4.15; 95% CI: 1.12–15.78). More frequent durable complete renal responses were achieved with Isa-Pd (31.3%) versus Pd (19.0%; OR: 1.93; 95% CI: 0.45–9.82). A minor renal response was observed in one patient receiving Isa-Pd. Among responders, the median time to renal response was 3.4 weeks for patients treated with Isa-Pd and 7.3 weeks for patients treated with Pd.

Fewer patients progressed to end-stage renal disease (eGFR <15 ml/min/1.73 m²) during treatment with Isa-Pd compared with Pd (2.9 vs 7.9%). Among patients with moderate renal impairment at baseline, renal function worsened to severe renal impairment or end-stage renal disease in 22.6% of patients treated with Isa-Pd and 34.8% of patients treated with Pd (OR: 0.55; 95% CI: 0.20–1.45).

All patients with renal impairment experienced AEs, whereas 98.8 and 96.8% of patients without renal impairment experienced AEs in the Isa-Pd and Pd arms, respectively. There was a \geq 10% higher incidence of cardiac disorders and infections in patients with versus without renal impairment in the Isa-Pd arm. The incidence of grade \geq 3 AEs was similar for all patients treated with Isa-Pd, regardless of renal impairment status (90.7% with renal impairment vs 86% without renal impairment). IRs were more common in patients receiving Isa-Pd than in patients receiving Pd, affecting around a third of patients both with and without renal impairment. The incidence of AEs leading to definitive treatment discontinuation was higher in patients with renal impairment who received either Isa-Pd or Pd.

The pharmacokinetics of isatuximab were comparable between the renal subgroups, which was expected because monoclonal antibodies are not renally excreted. Therefore, there is no pharmacokinetics-based need for dose adjustment in patients with renal impairment. The results of this ICARIA-MM subgroup analysis provide evidence of improvement in renal function with Isa-Pd treatment. Results from the EQUULEUS study investigating the combination of daratumumab, pomalidomide and dexamethasone demonstrated that patients with and without renal impairment exhibited a similar ORR (58.1 and 61.1%, respectively), which was also comparable with that observed in the entire cohort (60.2%) [49]. To be included, patients needed a calculated creatinine clearance \geq 45 ml/min/1.73 m²; therefore, these patients had less renal impairment compared with those included in the ICARIA-MM. There are currently no studies reporting renal response in patients with MM treated with daratumumab.

ICARIA-MM subgroup: elderly patients

MM is a neoplastic disease which typically affects elderly patients. MM is most frequently diagnosed in people aged 65–74 years, with a median age at diagnosis of 69 years [61,62]. Elderly patients comprise a heterogeneous population with the potential for various co-morbidities, reduced functional status and increased risk of frailty. Furthermore, due to restrictive trial entry criteria in most clinical trials, the majority of very elderly patients are excluded, and the candidates enrolled in trials aged \geq 75 years tend to be unusually fit and are not representative of the typical elderly patient [63].

Due to its prognostic relevance, age (<75 vs ≥ 75 years) was one of the stratification factors of ICARIA-MM. A prespecified subgroup analysis comparing efficacy and safety in patients aged ≥ 75 years, 65–74 years and less than

65 years was conducted [39,64]. The median age of patients was 68.0 years in the Isa-Pd arm and 66.0 years in the Pd arm. In the Isa-Pd and Pd arms, there were 54 (35.1%) and 70 (45.8%) patients younger than 65 years of age, 68 (44.2%) and 54 (35.3%) patients 65–74 years of age, and 32 (20.8%) and 29 (19.0%) patients \geq 75 years of age, respectively.

PFS was significantly improved with Isa-Pd versus Pd in all age groups; patients \geq 75 years of age had a median PFS of 11.4 versus 4.5 months, respectively (HR: 0.48; 95% CI: 0.24–0.95). Similarly, in the Isa-Pd and Pd groups, patients 65–74 years of age had a PFS of 11.6 and 8.6 months (HR: 0.64; 95% CI: 0.39–1.06); in patients younger than 65 years of age, PFS was 11.5 versus 5.0 months, respectively (HR: 0.66; 95% CI: 0.40–1.07).

The ORRs in patients receiving Isa-Pd versus Pd by age group were as follows: 53.1 and 31.0% in patients aged \geq 75 years (OR: 2.52; 95% CI: 0.79–8.26); 64.7 and 38.9% in patients aged 65–74 years (OR: 2.88; 95% CI: 1.29–6.46); and 59.3 and 34.3% in patients aged younger than 65 years (OR: 2.79; 95% CI: 1.26–6.20). The proportion of patients achieving at least a VGPR or better by age in patients receiving Isa-Pd versus Pd was 31.3 versus 0% in those aged \geq 75 years (OR not calculated); 32.4 versus 13.0% in those aged 65–74 years (OR: 3.21; 95% CI: 1.17–9.70); and 31.5 versus 8.6% in those aged younger than 65 years (OR: 4.90; 95% CI: 1.64–16.35). Of eight patients with negative MRD at 10⁻⁵ in the Isa-Pd arm, two patients were \geq 75 years old.

In the Isa-Pd arm, nearly all patients reported AEs of any grade, with an incidence of 100% among patients aged \geq 75 years, 100% among those aged 65–74 years and 98.1% among those aged younger than 65 years. There were more grade \geq 3 AEs with Isa-Pd in patients aged \geq 75 years (93.8%) compared with patients younger than 65 years of age (85.2%), with a similar trend observed in the Pd arm (75.0 and 64.7%, respectively). There were also more treatment discontinuations due to AEs in patients \geq 75 versus <65 years of age who received Isa-Pd (15.6 and 7.4%, respectively) and Pd (14.3 and 10.3%, respectively).

There was a higher incidence of SAEs in patients \geq 75 versus less than 65 years of age in both arms (Isa-Pd, 68.8 and 57.4%; Pd, 57.1 and 47.1%, respectively). The incidence of AEs with fatal outcome was lower in patients aged \geq 75 years who received Isa-Pd (6.3%) than in patients aged less than 65 years (11.1%), whereas the opposite trend was observed with Pd (14.3 vs 5.9%). TEAEs were reported in 98.1 versus 97.1% of patients aged less than 65 years who received Isa-Pd versus Pd, respectively. In those who received Isa-Pd, the most common nonhematologic AE of any grade was IR, regardless of age group. There was a trend toward fewer IRs in patients aged \geq 75 years (28.1%) compared with patients aged 65–74 years (36.4%) or less than 65 years (42.6%).

In the Isa-Pd arm, AEs with the greatest difference in incidence for patients aged \geq 75 versus less than 65 years were IRs (28.1 vs 42.6%) and acute kidney injury (15.6 vs 1.9%). A similar difference in the incidence of acute kidney injury was seen among patients treated with Pd (10.7 vs 5.9%). In both treatment groups, grade \geq 3 AEs were more frequent in patients aged \geq 75 years than in those aged less than 65 years (Isa-Pd, 93.8 vs 85.2%; Pd, 75.0 vs 64.7%). The most common grade \geq 3 nonhematologic AE was pneumonia, regardless of patient age or treatment group. Grade 3–4 neutropenia and thrombocytopenia, based on laboratory results, were more common with Isa-Pd than Pd, regardless of patient age group. Neutropenia and infections were reversible and manageable with appropriate supportive care (G-CSF/granulocyte-macrophage colony-stimulating factor and antibiotics, respectively). The incidence of grade 4 thrombocytopenia was higher following Isa-Pd treatment compared with Pd treatment for patients aged \geq 75 and 65–74 years, whereas the contrary was true for patients aged less than 65 years. No patient discontinued isatuximab due to neutropenia or thrombocytopenia.

Overall, the addition of isatuximab to Pd improved PFS, ORR and rates of VGPR or better in elderly patients, consistent with the benefit observed in the overall study population [30]. There was a consistent trend toward higher rates of SAEs and discontinuation due to AEs in elderly patients in both the Isa-Pd and Pd arms, but no increase in fatal AEs was observed in the Isa-Pd arm. In the EQUULEUS study, the combination of daratumumab, pomalidomide and dexamethasone demonstrated an ORR of 60%, a PFS of 8.8 months and a median OS of 17.5 months at a median follow-up of 13.1 months [49]. The ORR was 57.7% among patients aged less than 65 and 62.7% among those aged 65 years or older. In a subgroup analysis of the Phase III POLLUX and CASTOR studies, daratumumab, lenalidomide and dexamethasone led to significantly prolonged PFS in patients aged 65–74 years (HR: 0.25; 95% CI: 0.16–0.40) and in those aged 75 years or older (HR: 0.26; 95% CI: 0.10–0.65) [65]. The daratumumab combination also led to improved ORR and MRD-negative status in both age groups. Grade 3/4 AEs were more common in patients aged 75 or older compared with younger patients (86.2 vs 77.1%, respectively). The combination of daratumumab, bortezomib and dexamethasone also contributed to significantly prolonged PFS, VGPR or better, and MRD-negative status regardless of age, with more grade 3/4 AEs reported in patients aged 75 years or older (IES). The patients in ICARIA-MM were older

than patients in EQUULEUS, CASTOR or POLLUX, and in particular, there was a higher percentage of patients aged 75 years or older included in ICARIA-MM. The ICARIA-MM study demonstrated a benefit with isatuximab treatment leading to improved PFS and ORR, with no increase in fatal AEs in this older patient population.

ICARIA-MM subgroup: prior lines & refractory status

The number of previous lines of treatment that a patient has received is an important consideration in treatment sequencing [66]. It is, therefore, important to evaluate novel therapies for RRMM based on patients' prior therapies, and another prespecified subgroup analysis of ICARIA-MM was designed to evaluate the efficacy of isatuximab by prior lines of treatment and refractory status [40,67]. Furthermore, due to restrictive trial entry criteria in most clinical trials, refractory patients may be excluded, resulting in enrolled candidates that may not be representative of the typical patient with MM [58,59].

In the ICARIA-MM study, patients receiving Isa-Pd and Pd were heavily pretreated, with patients receiving a median of three prior lines of therapy (range: 2–11) [31]. Overall, 66.1% received 2–3 prior lines of therapy and 33.9% received greater than three prior lines of therapy. A total of 92.5% of patients were refractory to Len and 75.9% to a PI, 58.9% of patients were refractory to Len at last line of treatment before study entry and 71.0% of patients were double refractory (Len + PI). A subgroup analysis revealed that the median PFS among Len-refractory patients with Isa-Pd (n = 144) was 11.4 versus 5.6 months with Pd (n = 140; HR: 0.60; 95% CI: 0.43–0.82). Additionally, PFS was higher with Isa-Pd (n = 93) versus Pd (n = 88) for patients that were refractory to Len as their last line of treatment before study entry (11.6 vs 5.7 months; HR: 0.50; 95% CI: 0.34–0.76). Median PFS in PI-refractory patients with Isa-Pd (n = 118) was 11.4 versus 5.6 months in the control group (n = 115; HR: 0.58; 95% CI: 0.41–0.82). In double-refractory patients, median PFS was 11.2 months with Isa-Pd (n = 111) and 4.8 months with Pd (n = 107; HR: 0.58; 95% CI: 0.40–0.84). Finally, PFS was higher in the Isa-Pd arm compared with the Pd arm for patients who had received 2–3 prior lines of therapy (12.3 vs 7.8 months; HR: 0.59; 95% CI: 0.40–0.88) and greater than three prior lines of therapy (9.4 vs 4.3 months; HR: 0.59; 95% CI: 0.36–0.98).

Patients refractory to Len, PIs, Len-PI and Len at last line of treatment had a higher ORR with Isa-Pd than Pd. ORR in patients receiving Isa-Pd versus Pd, respectively, was 59.0 versus 31.4% in Len-refractory patients; 60.2 versus 32.2% in PI-refractory patients; and 58.6 and 29.9% in double-refractory patients. The proportion of Len-refractory patients achieving VGPR or better was also higher with Isa-Pd versus Pd. Among the patients who received Isa-Pd, 30.6% (95% CI: 0.23–0.39) achieved VGPR or better compared with 7.1% (95% CI: 0.03–0.13) of patients who received Pd. Comparable results were also seen in PI-refractory patients, with 30.5% (95% CI: 0.22–0.40) versus 7.8% (95% CI: 0.04–0.14); in double refractory patients, with 29.7% (95% CI: 0.21–0.39) versus 8.4% (95% CI: 0.04–0.15); and in patients who were refractory to Len at last line of treatment, with 32.3% (95% CI: 0.23–0.43) versus 4.5% (95% CI: 0.01–0.11) achieving VGPR or better rate in the Isa-Pd versus the Pd arm, respectively.

Because refractory status reflects the treatment resistance of the myeloma clone, it should not impact systemic safety; however, prior lines of therapy reflect the extent of treatment exposure, which may influence safety. Therefore, safety for patients treated with Isa-Pd and Pd based on prior therapy lines are described below.

Among patients who received two prior lines of therapy, TEAEs were reported by 45 (100%) patients treated with Isa-Pd and by 43 (97.7%) patients treated with Pd. Grade \geq 3 TEAEs and treatment-emergent SAEs were more frequently reported in patients treated with Isa-Pd (86.7 and 60.0%) compared with Pd (63.6 and 47.7%). Fewer patients receiving Isa-Pd experienced TEAEs resulting in treatment discontinuation compared with patients receiving Pd (2.2 vs 6.8%).

Similar results were observed in patients receiving ≥ 3 prior lines of therapy. Grade ≥ 3 TEAEs and treatmentemergent SAEs were more frequently reported in patients treated with Isa-Pd (86.9 and 62.6%, respectively) compared with Pd (73.3 and 56.2%, respectively). Fewer patients receiving Isa-Pd experienced TEAEs, resulting in treatment discontinuation compared with patients receiving Pd (9.3 vs 15.2%, respectively).

The most common TEAE of any grade following isatuximab treatment was infection, affecting patients who had received 2 and \geq 3 prior lines of therapy similarly (80.4 vs 82.2%, respectively), which was similar to the results reported for the overall ICARIA-MM population [31]. Patients receiving two prior lines of therapy experienced more IRs following Isa-Pd treatment compared with patients who had received \geq 3 prior lines of therapy (48.9 vs 31.8%).

Overall, the addition of isatuximab to pomalidomide and dexamethasone improved PFS and ORR regardless of number of prior lines of therapy and in patients that were refractory to Len, refractory to Len at last line, and double refractory, consistent with the benefit observed in the overall ICARIA-MM population. Although cross-trial comparisons must be approached with caution due to differences in patient populations, results from the EQUULEUS study demonstrated a 58% ORR among double-refractory patients treated with the combination of daratumumab, pomalidomide and dexamethasone [49]. Additionally, 52% of the 103 patients had received more than three prior lines of therapy and the ORR for this subgroup was not significantly different from that of the whole cohort. The primary analysis of APOLLO, an ongoing Phase III trial of daratumumab plus pomalidomide and dexamethasone, showed a median PFS of 12.4 months (HR: 0.63) and an ORR of 69% in patients receiving daratumumab, pomalidomide and dexamethasone after a median follow-up of 16.9 months [68]. Of note, the median number of prior therapy lines was lower in APOLLO than in ICARIA-MM (2 vs 3) and the percentage of patients refractory to Len (79 vs 94%), a PI (47 vs 77%) or Len-PI (42 vs 72%) was also lower in APOLLO (daratumumab/pomalidomide/dexamethasone) than in ICARIA-MM (Isa-Pd).

Conclusion

The addition of isatuximab to pomalidomide and dexamethasone led to more frequent durable complete renal responses and improved PFS and disease response rates across different subgroups of patients with RRMM, independent of the high-risk cytogenetic status, renal impairment, age of patients or number of prior lines of treatment or refractory status. Treatment with Isa-Pd was able to overcome the adverse risk factors of age, refractory status and number of prior lines, resulting in similar PFS compared with the overall ICARIA-MM population.

Future perspective

Triplet therapy has become the standard treatment option for patients with RRMM, with a focus on anti-CD38 monoclonal antibodies as part of these regimens. Although recent treatment combinations have contributed to substantial improvements in survival of patients with MM, these benefits are not always realized among patients with poor prognostic criteria, including those with high-risk cytogenetics, those who are refractory to prior lines of treatment, those who have renal impairment, and those who are elderly, highlighting a need for improved and novel therapeutic strategies. Results from the ICARIA-MM trial demonstrated improved PFS and disease response rates, regardless of prognostics, with the addition of isatuximab to pomalidomide and dexamethasone making Isa-Pd an important treatment option for patients with RRMM [69]. As additional combinations and therapies, including CAR T-cell therapy and bispecific antibodies, are investigated for the treatment of patients with RRMM, subgroups with poor prognostic factors must be considered and included to help mitigate the unmet need for this patient population.

Executive summary

Multiple myeloma

- Multiple myeloma (MM) is the second most frequent hematologic malignancy. Currently, the majority of patients with MM are not cured.
- Patients with MM continuously relapse over time, highlighting a need for new therapeutic approaches, particularly among subgroups of patients who have a poor prognosis, including those with high-risk cytogenetics, those who are refractory to prior lines of treatment, those who have renal impairment and those who are elderly. Isatuximab

- Isatuximab is a monoclonal antibody that binds to a specific epitope of CD38, resulting in antibody-dependent cell-mediated cytotoxicity and exerting synergistic effects in combination with pomalidomide.
- Based on the Phase III ICARIA-MM study, isatuximab (Sarclisa[®]) is approved in a number of countries in combination with pomalidomide and dexamethasone for the treatment of adult patients with relapsed and refractory MM (RRMM) who have received at least two prior therapies, including lenalidomide and a proteasome inhibitor. Based on the Phase III IKEMA study, isatuximab in combination with carfilzomib and dexamethasone is also approved in the USA for the treatment of adult patients with RRMM who have received one to three prior lines of therapy, and in the European Union for the treatment of adult patients with MM who have received at least one prior therapy.

ICARIA-MM

- ICARIA-MM was a prospective, randomized, open-label, Phase III trial, which was designed to compare treatment of isatuximab in combination with pomalidomide (P) and dexamethasone ([d], Isa-Pd) with Pd in patients with RRMM.
- Patients who received Isa-Pd exhibited prolonged progression-free survival (PFS), increased overall response rate (ORR) and longer median time to progression compared with patients who received Pd.

• There was consistent benefit in the overall ICARIA-MM population and the subgroup populations presented in this review article.

High-risk cytogenetics

- High-risk cytogenetic markers associated with reduced survival in MM include the presence of t(4;14), t(14;16), del (17p) and/or gain(1q21).
- The addition of isatuximab to pomalidomide and dexamethasone provides a consistent benefit over pomalidomide and dexamethasone therapy alone in the difficult-to-treat subgroup of patients with RRMM who have high-risk cytogenetics.

Renal impairment

- The addition of isatuximab to pomalidomide and dexamethasone led to improved PFS and ORR among patients with renal impairment.
- Renal function is improved with the addition of isatuximab to pomalidomide and dexamethasone in patients with RRMM.

Elderly patients

- Adding isatuximab to pomalidomide and dexamethasone led to improved PFS and ORR in elderly patients aged 65–74 years and ≥75 years.
- A trend toward higher rates of serious adverse events and discontinuation due to adverse events was observed with both Isa-Pd and Pd treatment; however, there was no increase in fatal adverse events with Isa-Pd treatment.
- Prior lines/refractory status
- In ICARIA-MM, 76% of patients were refractory to lenalidomide or a proteasome inhibitor (PI), 59% were refractory to lenalidomide at last line of treatment and 71% of patients were refractory to both lenalidomide and a PI.
- The addition of isatuximab to pomalidomide and dexamethasone improved PFS and ORR regardless of the number of prior lines of therapy and in patients who are refractory to lenalidomide and/or a PI.

Supplementary data

An infographic accompanies this paper and is included at the end of the references section in the PDF version. To view or download this infographic in your browser please click here: www.futuremedicine.com/doi/suppl/10.2217/fon-2021-0568

Author contributions

F Campana, the funder's clinical study director, was responsible for oversight of the ICARIA-MM study. PG Richardson was a coprimary investigator of this study. PG Richardson, SJ Harrison, S Bringhen, F Schjesvold, K Yong and MA Dimopoulos were investigators of the study and contributed to data acquisition. PG Richardson, F Campana and S Le-Guennec designed the study. S Le-Guennec, S Macé, and F Campana contributed to the data analysis and interpretation. All the authors revised this work for important intellectual content and assume responsibility for integrity of the data and the decision to submit for publication; had full access to the study data; and edited and reviewed manuscript drafts, and approved the final draft for submission.

Acknowledgments

The authors thank the participating patients and their families, and the study centers and investigators, for their contributions to the study. They thank Michel Attal from the Institut Universitaire du Cancer de Toulouse-Oncopole, Toulouse, France, who was a coprimary investigator of the ICARIA-MM study.

Financial & competing interests disclosure

The ICARIA-MM study was sponsored by Sanofi. PG Richardson reports research funding from Bristol-Myers Squibb, Celgene, Oncopeptides and Takeda; and honoraria from Celgene, Janssen, Karyopharm, Oncopeptides, Sanofi and Takeda. SJ Harrison reports consultancy/membership on an entity's Board of Directors or advisory committees for AbbVie, Amgen, Celgene, GSK, Janssen Cilag, Novartis, Roche/Genentech, Takeda, Haemalogix and Sanofi; honoraria from Amgen, Celgene, Janssen Cilag, Novartis, Roche/Genentech and Takeda; and research funding from Amgen, Celgene, GSK, Janssen Cilag, Novartis and Haemalogix. S Bringhen reports honoraria from Amgen, Bristol-Myers Squibb, Celgene and Janssen; membership on an entity's Board of Directors or advisory committees for Amgen, Celgene, Janssen and Karyopharm; and consultancy for Janssen and Takeda. F Schjesvold reports honoraria from Amgen, Celgene, Janssen and Karyopharm; and consultancy for Janssen and Takeda. MA Dimopoulos reports consultancy/membership on an entity's Board of Directors or advisory committees for Amgen, Celgene, Janssen and Karyopharm; and consultancy for Janssen and Takeda. MA Dimopoulos reports consultancy/membership on an entity's Board of Directors or advisory committees for Amgen, Celgene, Janssen and Karyopharm; and consultancy for Janssen and Takeda. MA Dimopoulos reports consultancy/membership on an entity's Board of Directors or advisory committees for Amgen, Celgene, Janssen and Karyopharm; and consultancy for Janssen and Takeda. MA Dimopoulos reports consultancy/membership on an entity's Board of Directors or advisory committees for Amgen, Celgene and Takeda. K Yong has nothing to disclose. F Campana, S Le-Guennec and S Macé are employed by Sanofi and may hold stock and/or stock options in the company. The authors have no other relevant affiliations or financial involvement with any organization or

entity with a financial interest in or financial conflict with the subject matter or materials discussed in the manuscript apart from those disclosed.

Medical writing support was provided by Camile Semighini Grubor and Erin Burns-Tidmore, of Elevate Medical Affairs, contracted by Sanofi Genzyme for publication support services.

Data sharing statement

Qualified researchers can request access to patient-level data and related study documents including the clinical study report, study protocol with any amendments, blank case report forms, statistical analysis plan and dataset specifications. Patient-level data will be anonymized, and study documents will be redacted to protect the privacy of trial participants. Further details on Sanofi's data-sharing criteria, eligible studies and process for requesting access are at: https://www.clinicalstudydatarequest.com.

Open access

This work is licensed under the Attribution-NonCommercial-NoDerivatives 4.0 Unported License. To view a copy of this license, visit http://creativecommons.org/licenses/by-nc-nd/4.0/

References

Papers of special note have been highlighted as: • of interest; •• of considerable interest

- 1. Fairfield H, Falank C, Avery L, Reagan MR. Multiple myeloma in the marrow: pathogenesis and treatments. *Ann. NY Acad. Sci.* 1364, 32–51 (2016).
- Mussap M, Merlini G. Pathogenesis of renal failure in multiple myeloma: any role of contrast media? *Biomed. Res. Int.* 2014, 167125 (2014).
- 3. Kazandjian D. Multiple myeloma epidemiology and survival: a unique malignancy. Semin. Oncol. 43(6), 676-681 (2016).
- 4. Rajkumar SV, Kumar S. Multiple myeloma: diagnosis and treatment. Mayo Clin. Proc. 91(1), 101-119 (2016).
- WHO. International Agency for Research on Cancer. Globocan 2020 Multiple Myeloma Fact Sheet (2020). https://gco.iarc.fr/today/data/factsheets/cancers/35-Multiple-myeloma-fact-sheet.pdf
- 6. Palumbo A, Anderson K. Multiple myeloma. N. Engl. J. Med. 364(11), 1046–1060 (2011).
- Cavo M, Gay F, Beksac M *et al.* Autologous haematopoietic stem-cell transplantation versus bortezomib-melphalan-prednisone, with or without bortezomib-lenalidomide-dexamethasone consolidation therapy, and lenalidomide maintenance for newly diagnosed multiple myeloma (EMN02/HO95): a multicentre, randomised, open-label, Phase III study. *Lancet Haematol.* 7(6), e456–e468 (2020).
- Attal M, Lauwers-Cances V, Hulin C et al. Lenalidomide, bortezomib, and dexamethasone with transplantation for myeloma. N. Engl. J. Med. 376(14), 1311–1320 (2017).
- 9. Kumar SK, Rajkumar SV, Dispenzieri A *et al.* Improved survival in multiple myeloma and the impact of novel therapies. *Blood* 111(5), 2516–2520 (2008).
- Paquin AR, Kumar SK, Buadi FK et al. Overall survival of transplant eligible patients with newly diagnosed multiple myeloma: comparative effectiveness analysis of modern induction regimens on outcome. Blood Cancer J. 8(12), 125 (2018).
- 11. Thorsteinsdottir S, Dickman PW, Landgren O *et al.* Dramatically improved survival in multiple myeloma patients in the recent decade: results from a Swedish population-based study. *Haematologica* 103(9), e412–e415 (2018).
- 12. Morandi F, Horenstein AL, Costa F, Giuliani N, Pistoia V, Malavasi F. CD38: a target for immunotherapeutic approaches in multiple myeloma. *Front. Immunol.* 9, 2722 (2018).
- 13. Affif S, Michael A, Lesokhin A. Immunotherapy: a new approach to treating multiple myeloma with daratumumab and elotuzumab. *Ann. Pharmacother.* 50(7), 555–568 (2016).
- 14. Sarclisa (isatuximab-irfc). Prescribing information (2020). http://products.sanofi.us/Sarclisa/sarclisa.pdf
- Jagannath S, Abonour R, Durie BGM et al. Heterogeneity of second-line treatment for patients with multiple myeloma in the Connect MM registry (2010–2016). Clin. Lymphoma Myeloma Leuk. 18(7), 480–485, e483 (2018).
- 16. Raab MS, Fink L, Schoen P *et al.* Evolution of multiple myeloma treatment practices in Europe from 2014 to 2016. *Br. J. Haematol.* 185(5), 981–984 (2019).
- 17. Song X, Cong Z, Wilson K. Real-world treatment patterns, comorbidities, and disease-related complications in patients with multiple myeloma in the United States. *Curr. Med. Res. Opin.* 32(1), 95–103 (2016).
- 18. Rajkumar SV. Multiple myeloma: 2016 update on diagnosis, risk-stratification, and management. Am. J. Hematol. 91(7), 719–734 (2016).
- 19. Kumar SK, Therneau TM, Gertz MA *et al.* Clinical course of patients with relapsed multiple myeloma. *Mayo Clin. Proc.* 79(7), 867–874 (2004).

- Biran N, Jagannath S, Chari A. Risk stratification in multiple myeloma, part 1: characterization of high-risk disease. *Clin. Adv. Hematol.* Oncol. 11(8), 489–503 (2013).
- 21. Deaglio S, Mehta K, Malavasi F. Human CD38: a (r)evolutionary story of enzymes and receptors. Leukemia Res. 25(1), 1–12 (2001).
- 22. Deaglio S, Vaisitti T, Billington R et al. CD38/CD19: a lipid raft-dependent signaling complex in human B cells. Blood 109(12), 5390–5398 (2007).
- Malavasi F, Deaglio S, Funaro A *et al.* Evolution and function of the ADP ribosyl cyclase/CD38 gene family in physiology and pathology. *Physiol. Rev.* 88(3), 841–886 (2008).
- 24. Martin TG, Corzo K, Chiron M *et al.* Therapeutic opportunities with pharmacological inhibition of CD38 with isatuximab. *Cells* 8(12), 1522 (2019).
- Deckert J, Wetzel MC, Bartle LM et al. SAR650984, a novel humanized CD38-targeting antibody, demonstrates potent antitumor activity in models of multiple myeloma and other CD38+ hematologic malignancies. Clin. Cancer Res. 20(17), 4574–4583 (2014).
- This initial study demonstrated the antitumor activity of isatuximab (SAR650984) in cell lines, primary multiple myeloma (MM) samples and MM xenograft models in immunodeficient mice. The findings warranted Phase I clinical trial investigation in patients with CD38+ B-cell malignancies.
- 26. Moreno L, Perez C, Zabaleta A *et al.* The mechanism of action of the anti-CD38 monoclonal antibody isatuximab in multiple myeloma. *Clin. Cancer Res.* 25(10), 3176–3187 (2019).
- 27. Jiang H, Acharya C, An G *et al.* SAR650984 directly induces multiple myeloma cell death via lysosomal-associated and apoptotic pathways, which is further enhanced by pomalidomide. *Leukemia* 30(2), 399–408 (2016).
- 28. Atanackovic D, Yousef S, Shorter C *et al. In vivo* vaccination effect in multiple myeloma patients treated with the monoclonal antibody isatuximab. *Leukemia* 34(1), 317–321 (2020).
- Moreau P, Dimopoulos MA, Yong K et al. Isatuximab plus carfilzomib/dexamethasone versus carfilzomib/dexamethasone in patients with relapsed/refractory multiple myeloma: IKEMA Phase III study design. Future Oncol. 16(2), 4347–4358 (2020).
- 30. Richardson PG, Attal M, Campana F *et al.* Isatuximab plus pomalidomide/dexamethasone versus pomalidomide/dexamethasone in relapsed/refractory multiple myeloma: ICARIA Phase III study design. *Future Oncol.* 14(11), 1035–1047 (2018).
- Attal M, Richardson PG, Rajkumar SV *et al.* Isatuximab plus pomalidomide and low-dose dexamethasone versus pomalidomide and low-dose dexamethasone in patients with relapsed and refractory multiple myeloma (ICARIA-MM): a randomised, multicentre, open-label, Phase III study. *Lancet* 394(10214), 2096–2107 (2019).
- •• This pivotal Phase III study demonstrated the safety and efficacy of isatuximab in combination with pomalidomide and dexamethasone in patients with relapsed/refractory MM.
- 32. Sanofi. European Commission approves Sarclisa[®] (isatuximab) for adults with relapsed and refractory multiple myeloma (2020). www.sanofi.com/en/media-room/press-releases/2020/2020-06-02-12-47-38
- This press release announced the approval of isatuximab by the European Commission for use in the 27 member states of the European Union, plus the UK, Iceland, Liechtenstein and Norway.
- Sanofi Japan (2020). Press release: www.sanofi.co.jp/-/media/Project/One-Sanofi-Web/Websites/Asia-Pacific/Sanofi-JP/Home/press-re leases/PDF/2020/200629-02.pdf?la=ja
- •• This press release announced the approval of isatuximab in Japan.
- 34. Ministry of Health for the Russian Federation. Sarclisa[®] (Isatuximab) Registration Certificate in the Russian Federation (2020). http://grls.rosminzdrav.ru/Grls_View_v2.aspx?routingGuid=c754e935-ad28-4c1d-a643-0564811a2a00&t=
- •• This press release announced the approval of isatuximab by the Russian Federation.
- Korean Ministry of Food and Drug Safety. Sarclisa[®] (isatuximab) approval by the Korean Ministry of Food & Drug Safety (2020). https://nedrug.mfds.go.kr/pbp/CCBBB01/getItemDetail?itemSeq=202008401
- •• This press release announced the approval of isatuximab by the Korean Ministry of Food & Drug Safety.
- European Medicines Agency (EMA). Medicines. Sarclisa (2021). www.ema.europa.eu/en/medicines/human/summaries-opinion/sarclisa-0
- Harrison SJ, Perrot A, Alegre A et al. Subgroup analysis of ICARIA-MM in relapsed/refractory multiple myeloma patients with high-risk cytogenetics. Br. J. Haematol. 194(1), 120–131 (2021).
- This subgroup analysis of the Phase III ICARIA-MM study demonstrated that the addition of isatuximab to pomalidomide and dexamethasone provides a consistent benefit over pomalidomide and dexamethasone therapy alone in the difficult-to-treat subgroup of patients with relapsed/refractory MM who have high-risk cytogenetics.
- Dimopoulos MA, Leleu X, Moreau P et al. Isatuximab plus pomalidomide and dexamethasone in relapsed/refractory multiple myeloma patients with renal impairment: ICARIA-MM subgroup analysis. *Leukemia* 35(2), 562–572 (2021).
- •• This subgroup analysis of the Phase III ICARIA-MM study demonstrated that the addition of isatuximab to pomalidomide and dexamethasone led to improved progression-free survival (PFS) and overall response rate (ORR) among patients with renal impairment.

- 39. Schjesvold FH, Richardson PG, Facon T *et al.* Isatuximab plus pomalidomide and dexamethasone in elderly patients with relapsed/refractory multiple myeloma: ICARIA-MM subgroup analysis. *Haematologica* 106(4), 1182–1187 (2020).
- •• This subgroup analysis of the Phase III ICARIA-MM study demonstrated that the addition of isatuximab to pomalidomide and dexamethasone led to improved PFS and ORR among elderly patients aged 65–74 years and ≥75 years, with no increase in fatal adverse events.
- Bringhen S, Pour L, Vorobyev V et al. Isatuximab plus pomalidomide and dexamethasone in patients with relapsed/refractory multiple myeloma according to prior lines of treatment and refractory status: ICARIA-MM subgroup analysis. *Leuk. Res.* 104, 106576 (2021).
- •• This subgroup analysis of the Phase III ICARIA-MM study demonstrated that the addition of isatuximab to pomalidomide and dexamethasone led to improved PFS and ORR regardless of the number of prior lines of therapy and in patients who are refractory to lenalidomide and/or a proteasome inhibitor.
- Harrison SJ, Richardson PG, Alegre A et al. Efficacy of isatuximab/pomalidomide/dexamethasone in relapsed/refractory multiple myeloma: ICARIA-MM high-risk cytogenetics subgroup analysis. Clin. Lymphoma Myeloma Leuk. 19(10), e33 (2019).
- 42. Rasche L, Kortum KM, Raab MS, Weinhold N. The impact of tumor heterogeneity on diagnostics and novel therapeutic strategies in multiple myeloma. *Int. J. Mol. Sci.* 20(5), 1248 (2019).
- 43. Kalff A, Spencer A. The t(4;14) translocation and FGFR3 overexpression in multiple myeloma: prognostic implications and current clinical strategies. *Blood Cancer J.* 2, e89 (2012).
- Palumbo A, Avet-Loiseau H, Oliva S et al. Revised International Staging System for multiple myeloma: a report from International Myeloma Working Group. J. Clin. Oncol. 33(26), 2863–2869 (2015).
- Sonneveld P, Avet-Loiseau H, Lonial S et al. Treatment of multiple myeloma with high-risk cytogenetics: a consensus of the International Myeloma Working Group. Blood 127(24), 2955–2962 (2016).
- Fonseca R, Bergsagel PL, Drach J et al. International Myeloma Working Group molecular classification of multiple myeloma: spotlight review. Leukemia 23(12), 2210–2221 (2009).
- Kaufman JL, Dimopoulos MA, White D et al. Daratumumab, lenalidomide, and dexamethasone in relapsed/refractory myeloma: a cytogenetic subgroup analysis of POLLUX. Blood Cancer J. 10(11), 111 (2020).
- Weisel K, Spencer A, Lentzsch S et al. Daratumumab, bortezomib, and dexamethasone in relapsed or refractory multiple myeloma: subgroup analysis of CASTOR based on cytogenetic risk. J. Hematol. Oncol. 13(1), 115 (2020).
- 49. Chari A, Suvannasankha A, Fay JW *et al.* Daratumumab plus pomalidomide and dexamethasone in relapsed and/or refractory multiple myeloma. *Blood* 130(8), 974–981 (2017).
- Lonial S, Dimopoulos M, Palumbo A *et al.* Elotuzumab therapy for relapsed or refractory multiple myeloma. *N. Engl. J. Med.* 373(7), 621–631 (2015).
- Dimopoulos MA, Dytfeld D, Grosicki S et al. Elotuzumab plus pomalidomide and dexamethasone for multiple myeloma. N. Engl. J. Med. 379(19), 1811–1822 (2018).
- 52. Premkumar V, Pan S, Lentzsch S, Bhutani D. Use of daratumumab in high risk multiple myeloma: a meta-analusis. *eJHaem* 1, 267–271 (2020).
- Mohan M, Weinhold N, Schinke C et al. Daratumumab in high-risk relapsed/refractory multiple myeloma patients: adverse effect of chromosome 1q21 gain/amplification and GEP70 status on outcome. Br. J. Haematol. 189(1), 67–71 (2020).
- 54. Clark AD, Shetty A, Soutar R. Renal failure and multiple myeloma: pathogenesis and treatment of renal failure and management of underlying myeloma. *Blood Rev.* 13(2), 79–90 (1999).
- 55. Gonsalves WI, Leung N, Rajkumar SV *et al.* Improvement in renal function and its impact on survival in patients with newly diagnosed multiple myeloma. *Blood Cancer J.* 5(3), e296–e296 (2015).
- Eleutherakis-Papaiakovou V, Bamias A, Gika D et al. Renal failure in multiple myeloma: incidence, correlations, and prognostic significance. Leuk. Lymphoma 48(2), 337–341 (2007).
- 57. Dimopoulos MA, Oriol A, Nahi H et al. Daratumumab, lenalidomide, and dexamethasone for multiple myeloma. N. Engl. J. Med. 375(14), 1319–1331 (2016).
- Stewart AK, Rajkumar SV, Dimopoulos MA *et al.* Carfilzomib, lenalidomide, and dexamethasone for relapsed multiple myeloma. *N. Engl. J. Med.* 372(2), 142–152 (2015).
- 59. Moreau P, Masszi T, Grzasko N *et al.* Oral ixazomib, lenalidomide, and dexamethasone for multiple myeloma. *N. Engl. J. Med.* 374(17), 1621–1634 (2016).
- Dimopoulos MA, Sonneveld P, Leung N et al. International Myeloma Working Group recommendations for the diagnosis and management of myeloma-related renal impairment. J. Clin. Oncol. 34(13), 1544–1557 (2016).
- 61. Madan S, Kumar S. Current treatment options for elderly patients with multiple myeloma: clinical impact of novel agents. *Therapy* 8(4), 415–429 (2011).
- 62. Howlader N, Noone A, Krapcho M et al. SEER cancer statistics review, 1975–2014 (2017). https://seer.cancer-gov/csr/1975-2014/

- 63. Willan J, Eyre TA, Sharpley F, Watson C, King AJ, Ramasamy K. Multiple myeloma in the very elderly patient: challenges and solutions. *Clin. Interv. Aging* 11, 423–435 (2016).
- 64. Schjesvold FH, Richardson PG, Facon T *et al.* Isatuximab plus pomalidomide and dexamethasone in elderly patients with relapsed/refractory multiple myeloma: ICARIA-MM subgroup analysis. *Haematologica* 106(4), 1182–1187 (2021).
- Mateos MV, Spencer A, Nooka AK *et al.* Daratumumab-based regimens are highly effective and well tolerated in relapsed or refractory multiple myeloma regardless of patient age: subgroup analysis of the Phase III CASTOR and POLLUX studies. *Haematologica* 105(2), 468–477 (2020).
- Laubach J, Garderet L, Mahindra A *et al.* Management of relapsed multiple myeloma: recommendations of the International Myeloma Working Group. *Leukemia* 30(5), 1005–1017 (2016).
- 67. Bringhen S, Attal M, Pour L et al. ICARIA-MM study: efficacy analysis according to prior lines of treatment. Clin. Lymphoma Myeloma Leuk. 19(10), e245–e246 (2019).
- Dimopoulos MA, Terpos E, Boccadoro M *et al.* Daratumumab plus pomalidomide and dexamethasone versus pomalidomide and dexamethasone alone in previously treated multiple myeloma (APOLLO): an open-label, randomised, Phase III trial. *Lancet Oncol.* 22(6), 801–812 (2021).
- 69. Richardson PG, San Miguel JF, Moreau P *et al.* Interpreting clinical trial data in multiple myeloma: translating findings to the real-world setting. *Blood Cancer J.* 8(11), 109 (2018).